Author’s response to reviews

Title: Clinical parameters and outcomes of necrotizing soft tissue infections secondary to gastrointestinal fistulas

Authors:

Kun Guo (guokunmedical@126.com)
Jianan Ren (jiananr@gmail.com)
Gefei Wang (Wgfwang@gmail.com)
Guosheng Gu (guguoshengde@163.com)
Xiuwen Wu (lygwxxw@163.com)
Zhiwu Hong (15951990306@163.com)
Huajian Ren (rhj1288@163.com)
Tao Zheng (jefferyzheng@yeah.net)
Weijie Li (jie00823@163.com)
Miao Fang (1714479860@qq.com)
Peter Nthumba (nthumba@gmail.com)
Jieshou Li (doc.wgf@gmail.com)

Version: 1 Date: 28 Oct 2018

Author’s response to reviews:

Dear editor,

We have sent the patient’s written informed consent to your email address (The patient's name is Ji Jun). The corresponding modification have been completed in the manuscript, please review again.

Thank you very much!